Author’s response to reviews

Title: Are variations in rates of attending cultural activities associated with population health in the United States?

Authors:

Anna V Wilkinson (awilkins@mdanderson.org)
Andrew J Waters (ajwaters@mdanderson.org)
Lars O Bygren (lars-olov.bygren@prevnut.ki.se)
Alvin R Tarlov (atarlov@gmail.com)

Version: 7 Date: 9 July 2007

Author’s response to reviews:

We have clarified (in the methods) that all aspects of the study received approval from the Institutional Review Board at the University of Chicago and that verbal consent was obtained from the study participants.